Accelerator Status Report
Run Schedule Jan. 14, 2009 to March 16, 2009
· Injector energy 65.64 MeV
· Linac Setup 583. 5 MeV

· Pass setup: 4/5/5
· The accelerator was affected by some box supplies and the vacuum at 1L04. Arc2, Arc4 and RSEP9A each tripped off a couple of times. EES replaced the faulty parts. The vacuum at 1L04 started to deteriorated Sunday and was getting worse Monday. Cryo module 1L04 is at about room temperature. The warm window of cavity 1L04-5 may have a leak. A pump was connected to it to help pump. The accelerator has been down since early Monday morning.
Hall A was changed to pass 4, Hall C to pass 5 and the Wien angle to -18 degrees. The team did a very good job. All was done within one hour (beam to Hall C dumplette).
· About 50 hours of beam time were lost last week due to vacuum problems, RF issues and a power glitch. 

· 1L06 insulating vacuum deteriorated due to a failed pump. The pump was replaced and the vacuum recovered (Thursday).
· The laser spot was moved to boost Hall A current.

· The Wien angle was changed to 30 degrees.

· DOG3E and DOG9E would not turn off. EES will troubleshoot it next week.

· The vacuum in 1L04 region started to deteriorate Friday. RF module 1L04 was cryo cycled Monday (3/2) and the vacuum was recovered.

· Sunday evening (3/1) a power glitch tripped off box supplies and the CHL.

· Arc9 GOFs were adjusted from -1.0 to -0.5mm to compensate for path length change.

· Trim magnet cycling: check what are not cycled and investigate why they are not.

Hall A
E06-014    4.7GeV     4 Pass
Measurement of Single Target-Spin Asymmetry
· The hall was changed back to pass 4. 

· The hall was changed to 5 pass. An Allsave file was used for the pass change.

· Hall A Moller procedure updated

Hall B
Eg1-DVCS    5.9 GeV    5pass

· No problem with Hall B beam

· Friday 2/27 the hall reported that something might be in its beam path.
A vertical line was seen from its target scans.

It turned out that IHA2H00 was not fully retracted.

Hall C

E07-003-SANE     5.9 GeV    5 Pass 
· The hall was changed back to pass 5. Hall C chicane was adjusted and beam is in “straight” mode. The chicane dipoles BE and BZ are off. The field of the target coil is parallel to beam so it does not bend beam (beam goes to the regular dump). 
· The hall was changed from pass 5 to pass 4 Thursday (2/26). An Allsave file was used for the pass change.
· There was a hot spot after the pass change: 3C19. The hot spot was due to beam striking in the region during the pass change. 
· Hall C chicane dipole communication problem was resolved by replacing the IP carrier on iochallmag.
